Author's response to reviews

Title: Ectopic insulin secreting neuroendocrine tumor of kidney with recurrent hypoglycemia

Authors:

Ramkumar S (ram.kumar@msn.com)
Atul Dhingra (dratuldhingra@gmail.com)
Ashraf M Ganie (Ashraf.endo@gmail.com)
Chandan J Das (docchandan17@gmail.com)
Amlesh Seth (amlesh.seth@gmail.com)
MC Sharma (sharmamehar@yahoo.co.in)
CS Bal (csbal@hotmail.com)
VP Jyotsna (vivekapjyotsna@gmail.com)

Version: 2 Date: 24 December 2013

Author's response to reviews: see over
Dear editors,

Thank you for your email. As per your email I have made the following changes in the manuscript and am submitting the revised version:

1. Authors contributions added.
2. Acknowledgements added
3. The Consent section located after the conclusion section
4. All the tables located after the references section
5. The Competing interest section has been placed after the Consent section.

Thanking you

Regards

Dr. Viveka P. Jyotsna
Associate Professor
Department of Endocrinology
All India Institute of Medical Sciences
New Delhi.

(Corresponding author: vivekajyotsna@gmail.com)